## RESEARCH

Annals of Intensive Care



# Which severe COVID-19 patients could benefit from high dose dexamethasone? A Bayesian post-hoc reanalysis of the COVIDICUS randomized clinical trial

Sylvie Chevret<sup>1</sup>, Lila Bouadma<sup>2,3</sup>, Claire Dupuis<sup>3,4</sup>, Charles Burdet<sup>3,5</sup> and Jean-François Timsit<sup>2,3\*</sup> on behalf of the COVIDICUS RCT group

## Abstract

Background The respective benefits of high and low doses of dexamethasone (DXM) in patients with severe acute respiratory syndrome coronavirus 2 (SARS-Cov2) and acute respiratory failure (ARF) are controversial, with two large triple-blind RCTs reaching very important difference in the effect-size. In the COVIDICUS trial, no evidence of additional benefit of high-dose dexamethasone (DXM20) was found. We aimed to explore whether some specific patient phenotypes could benefit from DXM20 compared to the standard of care 6 mg dose of DXM (DXMSoC).

Methods We performed a post hoc exploratory Bayesian analysis of 473 patients who received either DXMSoc or DXM20 in the COVIDICUS trial. The outcome was the 60 day mortality rate of DXM20 over DXMSoC, with treatment effect measured on the hazard ratio (HR) estimated from Cox model. Bayesian analyses allowed to compute the posterior probability of a more than trivial benefit (HR < 0.95), and that of a potential harm (HR > 1.05). Bayesian measures of interaction then quantified the probability of interaction (Pr Interact) that the HR of death differed across the subsets by 20%. Primary analyses used noninformative priors, centred on HR = 1.00. Sensitivity analyses used sceptical and enthusiastic priors, based on null (HR = 1.00) or benefit (HR = 0.95) effects.

Results Overall, the posterior probability of a more than trivial benefit and potential harm was 29.0 and 51.1%, respectively. There was some evidence of treatment by subset interaction (i) according to age (Pr Interact, 84%), with a 86.5% probability of benefit in patients aged below 70 compared to 22% in those aged above 70; (ii) according to the time since symptoms onset (Pr Interact, 99%), with a 99.9% probability of a more than trivial benefit when lower than 7 days compared to a < 0.1% probability when delayed by 7 days or more; and (iii) according to use of remdesivir (Pr Interact, 91%), with a 90.1% probability of benefit in patients receiving remdesivir compared to 19.1% in those who did not.

**Conclusions** In this exploratory post hoc Bayesian analysis, compared with standard-of-care DXM, high-dose DXM may benefit patients aged less than 70 years with severe ARF that occurred less than 7 days after symptoms onset. The use of remdesivir may also favour the benefit of DXM20. Further analysis is needed to confirm these findings.

\*Correspondence Jean-François Timsit

Jean-francois.timsit@bch.aphp.fr

Full list of author information is available at the end of the article



© The Author(s) 2023, corrected publication 2023. Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

*Trial registration*: NCT04344730, date of registration April 14, 2020 (https://clinicaltrials.gov/ct2/show/NCT04344730? term=NCT04344730&draw=2&rank=1); EudraCT: 2020-001457-43 (https://www.clinicaltrialsregister.eu/ctr-search/search?query=2020-001457-43).

**Keywords** SARS-Cov-2, Steroids, Dose, Sepsis, Acute respiratory failure, Remdesivir, Critical care, ARDS, Bayesian analysis, Interaction

## Background

In the early phases of the severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) pandemic, the benefit of low/intermediate doses of corticosteroids in patients with acute respiratory failure (ARF) was found by the RECOVERY trial [1], confirmed by a meta-analysis of ongoing studies [2]. Since then, dexamethasone (DXM) 6 mg has been included as a standard of care (SoC) in the management of SARS-CoV-2 ARF with oxygen requirements. However, despite 4 randomized clinical trials (RCTs) [3–6], including 2 triple-blind RCTs [3, 6], the benefit of high doses of corticosteroids (DXM 12 mg or more) compared to standard of care low/intermediate doses (DXMSoc) could not be demonstrated [7].

In the COVIDICUS multicentre randomized clinical trial [3], a total of 546 patients admitted to the intensive care unit (ICU) with SARS-CoV-2 acute respiratory hypoxemic failure (ARHF) were randomized 1:1 to either high-dose dexamethasone (DXM20, n=270) or DXM-SoC (n=276) (NCT04344730). Such a strategy failed to have any impact on 60 day mortality (DXM20, 25.9%, vs. DXMSoC, 26.8%) [3]. However, the absence of any treatment effect, on the whole, may represent a benefit in some patients and harm in others due to differential treatment effects on subpopulations [8]. This was recently pointed out in the critical care setting, possibly related to the inclusion in trials of too heterogeneous populations [9]. Thus, exploring the treatment effect across different subgroups within an overall nonsignificant trial could be of interest [10].

Evaluation of the heterogeneity of the treatment effect is an essential aspect of personalized medicine and patient-centred outcome research. Factors that allow us to identify individuals who are more likely than others to experience a favourable or unfavourable effect of treatment define "predictive" factors, different from "prognostic" factors, defined as those used to identify the likelihood of a clinical event such as progression or death in patients. In patients with severe COVID-19 admitted to the ICU, several subsets of interest have been reported in the literature suggesting a prognostic [11–15] or predictive impact of those subsets [1, 5, 16–19].

However, to determine whether a factor is potentially predictive, a formal assessment of an interaction between the factor and treatment group needs to be performed [20]. Indeed, as with overall clinical trial results, chance findings are possible when assessing subgroup results. To assess the existence of interactions, the traditional approach evaluates the data in each of the subgroups independently and then uses several statistical tests for interaction, such as that of Gail and Simon [21]. However, clinical trials are rarely powered to detect statistically significant interactions. Bayesian approaches [22] have been reported as a novel solution to identify subgroups towards the "personalized medicine" [23]. Rather than postulating hypotheses regarding the quantity of interest, their main advantage is transparently communicating information by giving direct probabilistic statements [24]. In the setting of multi-population trials, Millen et al. [25] have proposed Bayesian interaction measures, referring to a potential concern that the inferences in the overall population may be unduly influenced by the treatment effect in a subgroup of patients.

In this study, based on the COVIDICUS trial, we used a Bayesian framework to assess the predictive value of several subsets of interest on the benefit of high-dose DXM in SARS-CoV-2 ARHF patients admitted to ICUs.

## Methods

## The covidicus trial

the **COVIDICUS** Study participation in trial (NCT04344730), sponsored by Assistance Publique-Hôpitaux de Paris (Paris, France), was proposed to all consecutive COVID-19 adult patients admitted to participating French ICUs who met the eligibility criteria. Eligible patients were adults aged  $\geq$  18 years admitted to the ICU within the last 48 h for confirmed or highly suspected COVID-19 infection and with signs of AHRF (PaO2<70 mmHg or transcutaneous oxygen saturation (SpO2) < 90% on room air, tachypnea > 30/min, laboured breathing, respiratory distress, or need for oxygen flow  $\geq$  6 L/min) and who could receive any available treatment intended to treat SARS-CoV-2 infection.

It aimed to compare the benefit of high-dose dexamethasone (DXM20, 20 mg/d for 5 days, then 10 mg/d  $\times$  5 days) compared to the standard of care (DXMSoC), first based on placebo. On July 3, 2020, after the publication of the recovery trial [1], the COVIDICUS Scientific Committee prompted the study group to amend the study protocol to allow investigators to administer DXM up to 6 mg/d for 10 days to DXMSoC patients. It also addressed the question of oxygen support technique, further comparing continuous positive airway pressure (CPAP) or high-flow oxygen therapy (HFNO) vs. standard oxygen support in non-intubated patients; however, we only focused on the effect of dexamethasone in this study.

Inclusions ranged from April 2020 to January 2021 in 19 French ICUs. The primary outcome was the timeto-all causes of death at Day 60 in the intent-to-treat population.

The trial was conducted in accordance with the Declaration of Helsinki. Signed informed consent was obtained from all included patients. An emergency consent procedure with the patient's legal guardian or relatives was implemented for patients unable to consent.

## Patients

Of the 546 randomized patients, 73 were included before September 17, 2020, when the protocol was amended to switch the placebo control group to a low dose of dexamethasone. We included all 473 patients from the modified intention-to-treat (ITT) population who were enrolled thereafter and randomly allocated to either DXM high dose (DXM20, n=234) or low dose (DXM-SoC, n=239).

### Subsets of interest

We first considered four partitions of patients based on (i) age (<,>70 years), (ii) inflammatory status, defined at admission by either ferritin > 1000  $\mu$ g/L or CRP>100 mg/L, as previously reported [26]; (iii) time elapsed since the onset of COVID19 symptoms at admission, using 7 days as the threshold [1, 16, 27]; and (iv) fever (body temperature < 38 °C vs.  $\geq$  38 °C). Then, we also considered the effect according to values of CRP, ferritin and Ddimers, using the median value in the whole sample as the cut-off value (i.e., 135, 1120 and 940, respectively); for ferritin, we also considered the reported cut-off of 3150.29 µg/L, as used in [28]. We also considered the severity of the disease, as measured by the need of invasive mechanical ventilation (IMV) at study entry, or according to the median value of the SAPS2. Finally, we also tested the impact of the concomitant use of remdesivir.

#### Statistical analysis

The treatment effect was defined as the hazard ratio (HR) of 60 day mortality in randomized groups DXM20 (high dose) vs. DXMSoC (low/moderate dose). We used the following Cox proportional hazards model  $\lambda(t) = \lambda_0(t) exp(\alpha t + \beta x + \gamma tx)$ , where  $\lambda_0(t)$  represents the baseline hazard function, x denotes a binary covariate,  $\beta$  the regression coefficient corresponding to the covariate, *t* is a binary treatment indicator,  $\alpha$  represents the treatment effect for patients with x = 0, and  $\gamma$  is the regression coefficient corresponding to the treatmentby-covariate interaction; treatment effect for patients with x = 1 is thus given by  $\alpha + \beta + \gamma$ . The estimation of the regression coefficients in the Cox model was performed in a Bayesian framework, with baseline hazard function defined as a mixture of piecewise constant functions [29]. We considered a total number of knots K=3 and an equally spaced partition of the time axis from 0 to 60 which corresponds to the longest survival time observed. The posterior distribution of each parameter was obtained, with the derived posterior density of any linear combination of the parameters, quantifying the uncertainty of treatment effect in each subset [30]. As proposed by Harrell in COVID-19 trials [31], a more than trivial benefit or a more than trivial harm was measured using the cut-off threshold of 1.05 on the HR scale. Thus, the posterior probability of a more than trivial benefit (HR < 0.95) and a more than trivial harm (HR > 1.05) overall and in each subset given the available data was computed. Then, according to Millen, treatment-by-subset interaction was measured on the ratio of treatment effect in the subsets, with the computed posterior probability that the HR of death in subsets differs by at least 20% [25].

Prior scenario was set under a non-informative independent framework with a gaussian N (0, 0.001) for each regression coefficient and an independent gamma distributions, Ga (0.01, 0.01) for each piecewise baseline hazard. Sensitivity analyses used optimistic and sceptical priors, that is centred on a positive (HR = 0.95) or null (HR = 1.00) effects, as recommended [32]. We also used Bayesian beta-binomial models, with the prevalence of death within the first 60 days following randomisation as the parameter of interest and the relative risk (RR) of death as the measure of effect.

We used R (https://www.R-project.org/) and JAGS [33], a user-friendly, open-source, validated software suited for the application of Bayesian methods, for analysis. We ran each model for 1000 burn-in simulations, then the model was run for 50,000 additional simulations to keep one in 10 so that a proper thinning is done. Gelman and Rubin's convergence diagnostic [34] was computed. Trace plots of the sampled values

for each parameter in the chain appear overlapping one another and Gelman–Rubin values were very close to 1, which indicated that convergence has been achieved.

## Results

The distribution of the 473 enrolled patients across the different strata is reported in Table 1. Most of the time, two subsets of imbalanced sizes were distinguished, with the lowest subset representing 11-29% of the sample.

In the whole modified ITT population, there was no evidence of any effect overall, as illustrated by the Bayesian posterior median HR of 60 day mortality estimated at 0.947 (95% credibility interval, 0.66–1.37), close to the frequentist estimate of 0.947 (95% confidence interval, 0.66–1.36) illustrating the non-informative priors (Fig. 1). Sensitivity analyses based on sceptical or enthusiastic prior only very slightly modified these findings (Fig. 1C). The posterior probability of a more than trivial benefit (HR < 0.95) and a more than trivial harm (HR > 1.05) in the DXM20 group was 0.51 and 0.29, respectively.

Some evidence of a treatment-by-subset interaction, that is, heterogeneity of the treatment effect in some subsets, was suggested (Fig. 2, Table 2). First of all, this concerned the patient's age: indeed, there was a 99.9% probability that the DXM20 benefit differed by at least 20%, with some evidence of benefit for patients aged under 70 years (HR=0.68, 95% CrI 0.37-1.23, probability of benefit 86.5%, probability of harm 7.7%) while on the opposite some evidence of deleterious effect in those aged above 70 years (HR = 1.15, 95% CrI 0.71-1.83, probability of benefit 22%, probability of harm 64%) (Fig. 3A). Otherwise, high-dose DXM may have benefited patients with treatment onset within the first 7 days of infection (HR = 0.59, 95% CrI 0.46 - 0.75) while it was deleterious in those who were admitted later (HR=1.16, 95% CI 1.06-1.28) (Fig. 3B), or in those with high levels of ferritin, with a 99% probability of benefit when ferritin > 1120  $\mu$ g/L compared to 2.3% in those < 1120  $\mu$ g/L (Fig. 3C, Table 2). Close findings were observed, though erased, according to CRP, with some evidence of decreased effect of DXM20 in patients with CRP>135 (HR ratio of 1.43, 95% CrI 0.79-2.59). Similarly, patients with low levels of Ddimers appeared to have benefited from DXM20 (HR=0.63, 95% CrI 0.34-1.17) compared to those with high levels in whom the treatment appeared deleterious (HR = 1.52, 95% CrI 0.87-2.72) (HR ratio = 2.41, 95% CrI 1.38-4.29). Finally, we observed an 91% probability of interaction between remdesivir use and DXM20 effect on Day 60 mortality, where remdesivir use was associated with a 90% chance of possible benefit of DXM20 (HR=0.61, 95% CrI 0.29-1.18) compared to 19% in those who did not receive remdesivir (HR = 1.15, 95% CrI

Table 1	Characteristics	of patients	across treatme	nt groups and

|--|

ha

Baseline subsets at randomization         DXMSoC, $n = 239$ N (%)         DXM20, $n = 234N (%)           Age, years         -           < 70         144 (60.2)         135 (57.7)           \geq 70         95 (39.8)         99 (42.3)           Time since symptoms onset, days         -         -           < 7         55 (23.1)         52 (22.9)           \geq 7         183 (76.9)         175 (77.1)           Body Temperature, °C         -         -           < 38         186 (79.1)         182 (78.1)           \geq 38         49 (20.8)         51 (21.9)           Inflammation syndromea         -         -           No         26 (13.8)         23 (13.2)           Yes         162 (86.2)         151 (86.8)           Missing         51         60           Age < 60 and inflammation         114 (60.6)         122 (70.1)           Age < 60 and inflammation         144 (60.6)         122 (70.1)           Age < 60 and inflammation         23 (12.2)         13 (7.5)           Missing         51         60           CRP, mg/L         -         -           < 135         98 (49.7)         98 (50.0)           > 135$	
Age, years $< 70$ 144 (60.2)135 (57.7) $\geq 70$ 95 (39.8)99 (42.3)Time since symptoms onset, days $< 7$ 55 (23.1)52 (22.9) $\geq 7$ 183 (76.9)175 (77.1)Body Temperature, °C $< 38$ 186 (79.1)182 (78.1) $\geq 38$ 49 (20.8)51 (21.9)Inflarmation syndrome <sup>a</sup> $< 31 (1.2)$ 151 (86.8)No26 (13.8)23 (13.2)Yes162 (86.2)151 (86.8)Missing5160Age < 60 and no inflammation3 (1.6)10 (5.7)Age < 60 and inflammation48 (25.5)29 (16.7)Age < 60 and inflammation23 (12.2)13 (7.5)Missing5160CRP, mg/L $< 135$ 98 (49.7)98 (50.0) $\geq 135$ 99 (50.3)98 (50.0)Missing4238	ł
<70144 (60.2)135 (57.7) $\geq$ 7095 (39.8)99 (42.3)Time since symptoms onset, days $<$ 755 (23.1)52 (22.9) $\geq$ 7183 (76.9)175 (77.1)Body Temperature, °C $<$ 38186 (79.1)182 (78.1) $\geq$ 3849 (20.8)51 (21.9)Inflammation syndrome <sup>a</sup> No26 (13.8)23 (13.2)Yes162 (86.2)151 (86.8)Missing5160Age < 60 and no inflammation	
≥ 7095 (39.8)99 (42.3)Time since symptoms onset, days<7	
$< 7$ 55 (23.1)       52 (22.9) $\geq 7$ 183 (76.9)       175 (77.1)         Body Temperature, °C $< 38$ 186 (79.1)       182 (78.1) $\geq 38$ 49 (20.8)       51 (21.9)         Inflarmation syndrome <sup>a</sup> $< 26 (13.8)$ 23 (13.2)         Yes       162 (86.2)       151 (86.8)         Missing       51       60         Age < 60 and no inflammation*	
$< 7$ 55 (23.1)       52 (22.9) $\geq 7$ 183 (76.9)       175 (77.1)         Body Temperature, °C $< 38$ 186 (79.1)       182 (78.1) $\geq 38$ 49 (20.8)       51 (21.9)         Inflammation syndrome <sup>a</sup> 26 (13.8)       23 (13.2)         Yes       162 (86.2)       151 (86.8)         Missing       51       60         Age < 60 and no inflammation*	
≥ 7 Body Temperature, °C < 38 38 38 38 38 38 49 (20.8) 51 (21.9) 182 (78.1) 238 49 (20.8) 51 (21.9) 101 101 102 (86.2) 151 (86.8) 151 (86.8) 151 (86.8) 151 (86.8) 151 (86.8) 151 (86.8) 151 (86.8) 151 (86.8) 103 (5.7) 4ge < 60 and no inflammation 4ge < 60 and inflammation 4ge < 60 and inflammation 4ge < 60 and inflammation 4ge < 60 and no i	
800y Temperature, °C $< 38$ 186 (79.1)       182 (78.1) $\geq 38$ 49 (20.8)       51 (21.9)         Inflammation syndrome <sup>a</sup> 26 (13.8)       23 (13.2)         No       26 (13.8)       23 (13.2)         Yes       162 (86.2)       151 (86.8)         Missing       51       60         Age < 60 and no inflammation*	
<38	
≥ 38 49 (20.8) 51 (21.9)  Inflammation syndromea	
Inflammation syndrome <sup>a</sup> 26 (13.8)       23 (13.2)         No       26 (28.2)       151 (86.8)         Yes       162 (86.2)       151 (86.8)         Missing       51       60         Age < 60 and no inflammation*	
No26 (13.8)23 (13.2)Yes162 (86.2)151 (86.8)Missing5160Age < 60 and no inflammation*	
Yes162 (86.2)151 (86.8)Missing5160Age < 60 and no inflammation*	
Missing5160Age < 60 and no inflammation*	
Age < 60 and no inflammation*3 (1.6)10 (5.7)Age < 60 and inflammation	
Age ≥ 60 and inflammation114 (60.6)122 (70.1)Age < 60 and inflammation	
Age < 60 and inflammation48 (25.5)29 (16.7)Age $\geq$ 60 and no inflammation23 (12.2)13 (7.5)Missing5160CRP, mg/L $<$ $<$ < 135	
Age $\geq$ 60 and no inflammation23 (12.2)13 (7.5)Missing5160CRP, mg/L< 135	
Missing         51         60           CRP, mg/L <td></td>	
CRP, mg/L <135 98 (49.7) 98 (50.0) ≥135 99 (50.3) 98 (50.0) Missing 42 38	
< 135	
≥ 135 99 (50.3) 98 (50.0) Missing 42 38	
Missing 42 38	
Ferritin, µg/L	
<1120 76 (55.5) 61 (44.2)	
≥ 1120 61 (44.5) 77 (55.8)	
Missing 102 96	
Ferritin, µg/L	
< 3150.29 123 (89.8) 121 (87.7)	
≥ 3150.29 14 (10.2) 17 (12.3)	
Missing 102 96	
D-Dimers, ng/mL	
< 940 89 (46.4) 103 (53.9)	
≥ 940 103 (53.6) 88 (46.1)	
Missing 47 43	
Mode of oxygenation	
O2 face mask 48 (20.1) 46 (19.7)	
HFNO 48 (20.1) 41 (17.5)	
CPAP 47 (19.7) 50 (21.4)	
IMV 96 (40.1) 97 (41.4)	
Invasive mechanical ventilation	
No 143 (59.9) 137 (58.6)	
Yes 96 (40.1) 97 (41.4)	
SAPS II	
< 33 113 (50.9) 100 (47.6)	
≥ 33 109 (49.1) 110 (52.4)	
Other treatment of SARS-CoV-2	
Remdesivir 61 (25.5) 62 (26.5)	
IL6 1 (0.4) 4 (1.7)	

## Table 1 (continued)

CRP C reactive protein, O2 oxygen, HFNO High-Flow Nasal Oxygen, CPAP Continuous Positive Airway Pressure, IMV invasive mechanical ventilation, IL6 interleukin 6, SAPS II Simplified Acute Physiology Score 2

 $^a$  Defined as either ferritin level > 1000  $\mu g/L$  or CRP > 100 mg/L

0.75–1.75) (Fig. 3D). By contrast, there was no evidence of any treatment-by-subset interaction according to the fever (with a 0.58 posterior probability of HR differing by 20%) or SAPS2 (Probability of interaction of 0.67).

Sensitivity analyses are reported in Additional file. Modifying the prior in terms of baseline hazards or treatment effect did not affect the results (Additional



**Fig. 1** COVIDICUS Trial: Main trial outcome across the DXM randomized groups. Overall survival according to randomization (**a**) and posterior density of the hazard ratio (HR) of the 60-day mortality rate in the whole trial population based on a noninformative prior (**b**) or using either a sceptical or an enthusiastic prior (**c**)



Fig. 2 Looking for treatment by subset interactions in terms of hazard ratio (HR) of 60-day mortality. CRP C reactive protein, MV invasive mechanical ventilation, SAPS Simplified Acute Physiology Score

file 1: Table S1). When ignoring time to death by modelling the prevalence rather than the hazard of death in the first 60 days, detected interactions were also roughly similar. Nevertheless, previous observed heterogeneity in treatment effect across D-dimers or ferritin levels were erased using a beta-binomial model, likely due to the fact that all survival curves reached close 60 day estimates (Additional file 1: Fig S2). By contrast, three main interactions, with age, time to symptoms onset, and remdesivir, were confirmed.

Subsets	Posterior quantities						
	HR death, 95% credible interval	Pr benefit (HR < 0.95)	Pr harm (HR > 1.05)	HR ratio (95% credible int)	Pr interaction (HR ratio > 1.2)		
All	0.95 (0.66–1.37)	0.511	0.290				
Age, years							
<70	0.68 (0.37-1.23)	0.865	0.077	1.00	0.84		
≥70	1.15 (0.72–1.83)	0.22	0.64	1.69 (0.79–3.66)			
Days since symptoms							
<7	0.59 (0.46–0.75)	0.999	< 0.0001	1.00	0.99		
≥7	1.16 (1.06–1.28)	< 0.0001	0.981	1.99 (1.52–2.56)			
Body temperature, °C					0.58		
<38	0.96 (0.63–1.45)	0.487	0.335	1.00			
<u>≥</u> 38	0.93 (0.47-1.71)	0.526	0.357	0.97 (0.49–1.79)			
Inflammation <sup>a</sup>							
No	0.94 (0.30–2.79)	0.507	0.422	1.00	0.68		
Yes	0.91 (0.44–2.16)	0.541	0.366	0.97 (0.44–2.59)			
CRP, mg/L							
<135	0.93 (0.51–1.67)	0.533	0.340	1.00	0.76		
≥ 135	1.33 (0.74–2.38)	0.129	0.789	1.43 (0.79–2.59)			
Ferritin, µg/L							
<1120	1.87 (0.96–3.57)	0.023	0.957	1.00	0.99		
≥1120	0.42 (0.21-0.84)	0.99	< 0.001	0.22 (0.09–0.58)			
Ddimers, ng/mL							
< 940	0.63 (0.34–1.17)	0.906	0.051	1.00	0.99		
≥ 940	1.52 (0.87–2.72)	0.050	0.904	2.41 (1.38–4.29)			
IMV							
No	0.90 (0.53–1.43)	0.600	0.254	1.00	0.65		
Yes	1.02 (0.57–1.84)	0.401	0.463	1.14 (0.54–2.44)			
SAPS II							
<33	0.91 (0.43–1.88)	0.55	0.35	1.00	0.67		
≥ 33	0.97 (0.63–1.51)	0.46	0.36	1.07 (0.46–2.57)			
Remdesivir use							
No	1.15 (0.75–1.75)	0.191	0.660	1.00	0.91		
Yes	0.61 (0.29–1.18)	0.901	0.056	0.54 (0.25-1.03)			

Table 2 Bayesian estimation of treatment effects of DXM20 vs DXMSoc across baseline subsets, looking for treatment-by-subset interactions

In each subset, hazard ratio (HR) of death within the first 60 days (with 95% credibility intervals, CrI) were computed, with posterior probability of benefit or harm; then potential heterogeneity in effect across subsets was measured on the ratio of both HR

CRP C reactive protein, O2 oxygen, HFNO High-Flow Nasal Oxygen, CPAP Continuous Positive Airway Pressure, IMV invasive mechanical ventilation, IL6 interleukin 6, SAPS Simplified Acute Physiology Score, HR hazard ratio, Pr probability

 $^a$  Defined as either ferritin level > 1000  $\mu g/L$  or CRP > 100 mg/L

## Discussion

In the initial phase of the pandemic, large platform trials reported the benefit of corticosteroids (mainly low/intermediate doses) in SARS-CoV-2 AHRF [1, 2]. Recently, using all the available data included in a systematic review and meta-analysis [7], the Cochrane network concluded that systemic corticosteroids plus usual care probably reduces the number of deaths from any cause slightly, up to 30 days. No definite conclusion could be drawn about the number of deaths from any cause up to 120 days or on the optimal dose and duration of corticosteroids given to the patients.

In this post hoc analysis using a Bayesian approach, we first confirmed the absence of any treatment benefit on Day 60 mortality associated with high-dose DXM compared to low/intermediate doses of DXM (Table 2, Fig. 1). The Bayesian approach was considered, because it allows to reflect the uncertainty in treatment effect, illustrated



Fig. 3 Posterior density of the hazard ratio of death within 60 deaths in DXM20 over DXMSoc group, according to subsets. Subsets were defined by age (<, >70, Fig. 2a), by time since symptoms onset (<, >7 days, Fig. 2b) and by remdesivir use or not (Fig. 2c)

through posterior densities of the HR of death. Moreover, by contrast to frequentist approaches, further probabilistic statements could be derived from these distributions, such as the probability of benefit or harm, as well as the probability that the HR of death differs by 20% from one subset to another. This allowed us to quantify a 51% posterior probability of a more than trivial benefit and a 29% of a more than trivial arm of high-dose DXM in the whole sample. This result contradicts the Bayesian secondary analysis of the COVID-STEROIDS2 study [23]. The adjusted RR for 28 day mortality was 0.87 (95% CI 0.73–1.03), with probabilities of any benefit, clinically important benefit, and clinically important harm of 94.8, 80.7, and 0.9%, respectively. In addition to the difference in outcomes, many patient characteristics were very similar, i.e., time from symptoms onset and randomization (9 days in the median in both studies), IMV rate (17 and 21%), and age (mean 65 vs. 67 years). However, one-third of the patients from COVID-STEROIDS 2 were enrolled in low-income countries, with one-fifth randomized outside of the ICU. Moreover, remdesivir was used more frequently in COVID-STEROIDS2 than in COVIDICUS (62 vs 26%).

The potential benefit of remdesivir in intensive care patients particularly those with IMV/ECMO patient remained a matter of uncertainty [35, 36]. A recent individual patient data meta-analysis showed that remdesivir reduced mortality in patients hospitalized with COVID-19 who required no or conventional oxygen support, but was underpowered to evaluate patients who were ventilated when receiving remdesivir. The effect size of remdesivir in patients with more respiratory support and the cost-effectiveness of remdesivir remain to be further elucidated [37]. On the opposite a cohort study based on the PREMIER database including more than 40,000 patients found a very significant benefit of the use of remdesivir in ECMO/IMV patients [38]. The importance of the viral load (maximal at the early phase of the disease) might be more important than the intensity of oxygenation deterioration in selecting patients accessible to remdesivir therapy.

The potential heterogeneity in the DXM20 effect according to the use of remdesivir observed in our study may explain some of the discrepancies between the effects of high-dose DXM in COVID-STEROIDS2 and COVIDICUS. Indeed, in our study, there was a 90% chance of high-dose DXM benefit in patients receiving remdesivir compared to less than 20% in those who did not (Table 2). This is in agreement with the reported effect of corticosteroid therapy in delaying viral clearance: a small effect towards delayed time to viral clearance in young treated patients compared to young untreated patients was found in a large epidemiologic study [39]. In the same study, viral dynamics after hospitalization was an independent predictor of mortality (HR = 1.31, p < 10-3). Finally, a secondary analysis of the DISCOVERY study comparing remdesivir with the standard of care found that remdesivir use was associated with a small but significant increase in viral clearance [40]. We can, therefore, postulate that the potential benefit of high-dose DXM in the inflammatory process is offset by a detrimental effect on viral clearance. Remdesivir therapy may suppress this deleterious effect.

We also found that the posterior HR of death in DXM20 vs DXMSoc was 0.59 when the time from symptoms onset was <7 days, with an 99% probability of interaction between this delay and the DXM20 benefit; approximately one-quarter of both groups received remdesivir. In the RECOVERY study, a short delay between the first symptoms and randomization was associated with an insignificant impact of DXM-Soc6 on Day 28 mortality suggesting that 6 mg dose was not large enough [1]. Time from symptoms onset to corticosteroid administration did not impact the corticosteroid effect in the Outcomerea cohort [16]. However, the interaction between time from symptoms onset and high dose benefit was not found in the COVID-STEROID2 trial [6]. One possible hypothesis is that inflammation is more important in patients whose respiratory status rapidly deteriorates with a possible higher benefit of a high dose of corticosteroids. This is in line with the 99% posterior probability of a beneficial effect in patients with high inflammation as reflected by a high ferritin level (above 1120  $\mu$ g/L) compared to < 1% in those with lower levels. Unfortunately, the inflammation characteristics of the patients included in the COVID-STEROID2 study are not available.

The relationship between inflammatory reactions and corticosteroid effects was also suggested during the early phase of the pandemic, although based on observational data [27]. Using a latent class variable model, the authors found significant heterogeneity in the corticosteroid effect on mortality across inflammatory phenotypes, with corticosteroid exposure associated with decreased mortality in the hyperinflammatory phenotype and increased mortality in the hyperinflammatory phenotype [27]. Finally, an individualization of the corticosteroid dose based on the level of inflammation was suggested by a preliminary study but remains to be further evaluated [41]. The impact of hyperinflammation on the selection of patients who may benefit from high-dose DXM requires further study.

Our study has some limitations. We used a Bayesian modelling of the hazard of death. A multinormal model for the log hazard could have been used [30], but we choose to specify some model for the baseline hazard [42]. Thus, we used a piecewise exponential baseline hazard, with equally spaced knots while a random grid of timepoints could have been used [43]; nevertheless, the influence on posterior of prior specifications, including for the failure rates parameters, was evaluated and did not exhibit marked differences in results. The influence of inflammation status on the DXM20 effect differed according to the biomarker, and appeared more influenced by the ferritin level than by the CRP level, though discrepancies could rely on the choice of the cut-off points. We choose to rely on the literature or on the median value to limit overinterpretation of fishing. Moreover, the effect of Ddimers differed from that of ferritin; this could rely on sepsis-associated coagulopathy irrespective of the other inflammatory pathways [44]. Last, we focused our analyses of treatment by subset interaction to factors that have been evoked in previous studies of corticosteroid effects in SARS-CoV-2 ARF. Of course, other effect modifiers could have been considered. However, the use of Bayesian analyses to unmask possible effect modifiers is considered the best way to avoid enormous inflation of the risk of drawing erroneous conclusions [45]. Confirmed on all the sensitivity analyses, the potential interests of high dose dexamethasone when the delay from the first symptom is less than 7 days, in non-elderly patients or in combination with remdesivir ought to be further explored.

## Conclusions

Although no clear-cut evidence of an effect on 60-day mortality of high-dose corticosteroid therapy in patients with severe COVID-19 with ARF admitted to the ICU was observed, some subsets may benefit from such highdose steroids. Heterogeneity in effects according to age, time to ICU admission, and concomitant use of remdesivir was evidenced. This might be emphasized by the concurrent use of remdesivir in prompting viral clearance. As previously this result remained to be confirmed but argue for the use of remdesivir when high dose of DXM is decided based on the short delay between the first symptoms and ICU admission. These hypotheses need to be confirmed in further studies.

#### Abbreviations

ARF	Acute respiratory failure
Crl	Credible interval
DXM	Dexamethasone
HR	Hazard ratio
ICU	Intensive care unit
RCT	Randomized clinical trial
RR	Relative risk
SARS-CoV-2	Severe acute respiratory syndrome coronavirus 2

## Supplementary Information

The online version contains supplementary material available at https://doi. org/10.1186/s13613-023-01168-z.

Additional file 1: Table S1. Treatment effect on 60-day mortality across subsets. Figure S1. Sensitivity analyses using either Cox model with sceptical or enthusiastic priors, or beta-binomial models with noninformative priors. Figure S2. Estimated survival curves according to DXM group and time since disease onset (A), D-dimers (B) or ferritin (C) levels

Additional file 2. Location and role of the collaborators.

#### Acknowledgements

This study was only feasible because of the continuous implication of all the collaborators listed in the Additional file 2. This study has been labelled as a National Research Priority by the National Orientation Committee for Therapeutic Trials and other researches on Covid-19 (CAPNET). The investigators would like to acknowledge ANRS | Emerging infectious diseases for their scientific support, the French Ministry of Health and Prevention and the French Ministry of Higher Education, Research and Innovation for their funding and support.

#### Author contributions

SC analyzed and interpreted the patient data regarding the treatment effect and treatment-by-subset interactions. JFT initiated the study and was a major contributor in writing the manuscript. All authors read and approved the final manuscript.

#### Funding

The trial was funded by a grant from Programme Hospitalier de Recherche Clinique –PHRC 2020 (20–0388) (Ministry of Health). Vygon provided CPAP kits for some study centres.

#### Availability of data and materials

The datasets analyzed during the current study are not publicly available due to French regulation regarding trial data but are available from the corresponding author on reasonable request.

#### Declarations

#### Ethics approval and consent to participate

The trial protocol, initial and amended version, was approved by the institutional review board (Comité de Protection des Personnes Ile-de-France-XI, approval #20037-25957) and the French Health Authorities (dates, e-Methods). It was sponsored by Assistance Publique-Hôpitaux de Paris, registered on clinicaltrials.gov (NCT04344730) and Eudra-CT (2020-001457-43), and conducted in accordance with Helsinki Declaration. Consents were obtained in compliance with the French law for emergency inclusion, with signed informed consent obtained from conscious patients, and an emergency consent procedure with the patient's legal guardian/relatives implemented for those unable to consent.

#### **Consent for publication**

Not applicable.

#### **Competing interests**

J.F. Timsit reports lecture fees from Merck, Pfizer, Shionogi, Gilead, and Thermo Fisher and advisory board participation for Merck, Pfizer, Shionogi, Beckton-Dickinson, Aspen, and Gilead unrelated to the submitted work.

#### Author details

<sup>1</sup>ECSTRRA, UMR 1153, Saint Louis Hospital, University Paris Cité, Paris, France.
<sup>2</sup>Medical and Infectious Diseases ICU, APHP Bichat Hospital, 75018 Paris, France.
<sup>3</sup>Université Paris Cité, IAME, INSERM, UMR 1137, 75018 Paris, France.
<sup>4</sup>Intensive Care Unit, Gabriel Montpied Hospital, CHU de Clermont-Ferrand, 63000 Clermont-Ferrand, France.
<sup>5</sup>Epidemiology, Biostatistics and Clinical Research Department, AP-HP, Bichat Hospital, 75018 Paris, France.

Received: 8 May 2023 Accepted: 31 July 2023 Published: 27 August 2023

#### References

- The RECOVERY Collaborative Group. Dexamethasone in hospitalized patients with Covid-19. N Engl J Med. 2021;384(8):693–704. https://doi. org/10.1056/NEJMoa2021436.
- Sterne JAC, Murthy S, et al. Association between administration of systemic corticosteroids and mortality among critically III patients with COVID-19: a meta-analysis. JAMA. 2020;324(13):1330. https://doi.org/10. 1001/jama.2020.17023.
- Bouadma L, Mekontso-Dessap A, Burdet C, et al. High-dose dexamethasone and oxygen support strategies in intensive care unit patients with severe COVID-19 acute hypoxemic respiratory failure: the COVIDICUS randomized clinical trial. JAMA Intern Med. 2022. https://doi.org/10.1001/ jamainternmed.2022.2168.
- Maskin LP, Bonelli I, Olarte GL, et al. High-versus low-dose dexamethasone for the treatment of COVID-19-related acute respiratory distress syndrome: a multicenter and randomized open-label clinical trial. Intensive Care Crit Care Med. 2021. https://doi.org/10.1101/2021.09.15.21263597.
- Taboada M, Rodríguez N, Varela PM, et al. Effect of high versus low dose of dexamethasone on clinical worsening in patients hospitalised with moderate or severe COVID-19 pneumonia: an open-label, randomised clinical trial. Eur Respir J. 2022;60(2):2102518. https://doi.org/10.1183/13993003. 02518-2021.
- 6. Russell L, Uhre KR, et al. Effect of 12 mg vs 6 mg of dexamethasone on the number of days alive without life support in adults with COVID-19

and severe hypoxemia: the COVID STEROID 2 randomized trial. JAMA. 2021;326(18):1807. https://doi.org/10.1001/jama.2021.18295.

- Wagner C, Griesel M, Mikolajewska A, et al. Systemic corticosteroids for the treatment of COVID-19: equity-related analyses and update on evidence. Cochrane haematology group, ed. Cochrane Database Syst Rev. 2022. https://doi.org/10.1002/14651858.CD014963.pub2.
- Hemmings R. An overview of statistical and regulatory issues in the planning, analysis, and interpretation of subgroup analyses in confirmatory clinical trials. J Biopharm Stat. 2014;24(1):4–18. https://doi.org/10.1080/ 10543406.2013.856747.
- Granholm A, Munch MW, Myatra SN, et al. Dexamethasone 12 mg versus 6 mg for patients with COVID-19 and severe hypoxaemia: a pre-planned, secondary Bayesian analysis of the COVID STEROID 2 trial. Intensive Care Med. 2022;48(1):45–55. https://doi.org/10.1007/s00134-021-06573-1.
- Tanniou J, van der Tweel I, Teerenstra S, Roes KCB. Subgroup analyses in confirmatory clinical trials: time to be specific about their purposes. BMC Med Res Methodol. 2016;16(1):20. https://doi.org/10.1186/ s12874-016-0122-6.
- 11. Azoulay E, Fartoukh M, Darmon M, et al. Increased mortality in patients with severe SARS-CoV-2 infection admitted within seven days of disease onset. Intensive Care Med. 2020;46(9):1714–22. https://doi.org/10.1007/s00134-020-06202-3.
- Biccard BM, Gopalan PD, Miller M, et al. Patient care and clinical outcomes for patients with COVID-19 infection admitted to African high-care or intensive care units (ACCCOS): a multicentre, prospective, observational cohort study. Lancet. 2021;397(10288):1885–94. https://doi.org/10.1016/ S0140-6736(21)00441-4.
- Grasselli G, Zangrillo A, Zanella A, et al. Baseline characteristics and outcomes of 1591 patients infected with SARS-CoV-2 admitted to ICUs of the Lombardy Region, Italy. JAMA. 2020;323(16):1574. https://doi.org/10. 1001/jama.2020.5394.
- Gupta S, Hayek SS, Wang W, et al. Factors associated with death in critically III patients with coronavirus disease 2019 in the US. JAMA Intern Med. 2020;180(11):1436. https://doi.org/10.1001/jamainternmed.2020. 3596.
- Reyes LF, Murthy S, Garcia-Gallo E, et al. Respiratory support in patients with severe COVID-19 in the international severe acute respiratory and emerging infection (ISARIC) COVID-19 study: a prospective, multinational, observational study. Crit Care. 2022;26(1):276. https://doi.org/10.1186/ s13054-022-04155-1.
- Dupuis C, de Montmollin E, Buetti N, et al. Impact of early corticosteroids on 60-day mortality in critically ill patients with COVID-19: a multicenter cohort study of the OUTCOMEREA network. PLoS ONE. 2021;16(8):e0255644. https://doi.org/10.1371/journal.pone.0255644.
- Jeronimo CMP, Farias MEL, Val FFA, et al. Methylprednisolone as adjunctive therapy for patients hospitalized with coronavirus disease 2019 (COVID-19; Metcovid): a randomized, double-blind, phase IIb, placebocontrolled trial. Clin Infect Dis. 2021;72(9):e373–81. https://doi.org/10. 1093/cid/ciaa1177.
- Rubio-Rivas M, Ronda M, Padulles A, et al. Beneficial effect of corticosteroids in preventing mortality in patients receiving tocilizumab to treat severe COVID-19 illness. Int J Infect Dis. 2020;101:290–7. https://doi.org/ 10.1016/j.ijid.2020.09.1486.
- Torres A, Motos A, Cillóniz C, et al. Major candidate variables to guide personalised treatment with steroids in critically ill patients with COVID-19: CIBERESUCICOVID study. Intensive Care Med. 2022;48(7):850–64. https:// doi.org/10.1007/s00134-022-06726-w.
- Ballman KV. Biomarker: predictive or prognostic? J Clin Oncol Off J Am Soc Clin Oncol. 2015;33(33):3968–71. https://doi.org/10.1200/JCO.2015. 63.3651.
- Gail M, Simon R. Testing for qualitative interactions between treatment effects and patient subsets. Biometrics. 1985;41(2):361–72.
- Davis CE, Leffingwell DP. Empirical Bayes estimates of subgroup effects in clinical trials. Control Clin Trials. 1990;11(1):37–42. https://doi.org/10.1016/ 0197-2456(90)90030-6.
- Granholm A, Alhazzani W, Derde LPG, et al. Randomised clinical trials in critical care: past, present and future. Intensive Care Med. 2022;48(2):164– 78. https://doi.org/10.1007/s00134-021-06587-9.
- Ruberg SJ, Beckers F, Hemmings R, et al. Application of Bayesian approaches in drug development: starting a virtuous cycle. Nat Rev Drug Discov. 2023. https://doi.org/10.1038/s41573-023-00638-0.

- Millen BA, Dmitrienko A, Song G. Bayesian assessment of the influence and interaction conditions in multipopulation tailoring clinical trials. J Biopharm Stat. 2014;24(1):94–109. https://doi.org/10.1080/10543406. 2013.856025.
- Rubio-Rivas M, Mora-Luján JM, Montero A, et al. The use of corticosteroids or tocilizumab in COVID-19 Based on inflammatory markers. J Gen Intern Med. 2022;37(1):168–75. https://doi.org/10.1007/s11606-021-07146-0.
- 27. Sinha P, Furfaro D, Cummings MJ, et al. Latent class analysis reveals COVID-19–related acute respiratory distress syndrome subgroups with differential responses to corticosteroids. Am J Respir Crit Care Med. 2021;204(11):1274–85. https://doi.org/10.1164/rccm.202105-1302OC.
- Tom J, Bao M, Tsai L, et al. Prognostic and predictive biomarkers in patients with coronavirus disease 2019 treated with tocilizumab in a randomized controlled trial. Crit Care Med. 2021. https://doi.org/10.1097/ CCM.00000000005229.
- Goodman MS, Li Y, Tiwari RC. Detecting multiple change points in piecewise constant hazard functions. J Appl Stat. 2011;38(11):2523–32. https:// doi.org/10.1080/02664763.2011.559209.
- Simon R. Bayesian subset analysis: application to studying treatment-bygender interactions. Stat Med. 2002;21(19):2909–16. https://doi.org/10. 1002/sim.1295.
- 31. Harrell F, Lindsell C. Statistical Design and Analysis Plan for Sequential Parallel-Group RCT for COVID-19. http://hbiostat.org/proj/covid19/bayes plan.html
- Spiegelhalter DJ, Abrams KR, Myles JP. Bayesian approaches to clinical trials and health-care evaluation, vol. 13. Hoboken: John Wiley & Sons; 2004.
- Su YS. Package 'R2jags.' Published online August 22, 2015. https://CRAN.Rproject.org/package=R2jags
- Gelman A, Rubin DB. Inference from iterative simulation using multiple sequences. Stat Sci. 1992. https://doi.org/10.1214/ss/1177011136.
- 35. Ader F, Bouscambert-Duchamp M, Hites M, Peiffer-Smadja N, Poissy J, Belhadi D, Diallo A, Lê M-P, Peytavin G, Staub T, Greil R, Guedj J, Paiva J-A, Costagliola D, Yazdanpanah Y, Burdet C, Mentré F, Egle A, Greil R, Joannidis M, Lamprecht B, Altdorfer A, Belkhir L, Fraipont V, Hites M, Verschelden G, Aboab J, Ader F, Ait-Oufella H, Andrejak C, Andreu P, Argaud L, Bani-Sadr F, Benezit F, Blot M, Botelho-Nevers E, Bouadma L, Bouchaud O, Bougon D, Bouiller K, Bounes-Vardon F, Boutoille D, Boyer A, Bruel C, Cabié A, Canet E, Cazanave C, Chabartier C, Chirouze C, Clere-Jehl R, Courjon J, Crockett F, Danion F, Delbove A, Dellamonica J, Djossou F, Dubost C, Duvignaud A, Epaulard O, Epelboin L, Fartoukh M, Faure K, Faure E, Ferry T, Ficko C, Figueiredo S, Gaborit B, Gaci R, Gagneux-Brunon A, Gallien S, Garot D, Geri G, Gibot S, Goehringer F, Gousseff M, Gruson D, Hansmann Y, Hinschberger O, Jaureguiberry S, Jeanmichel V, Kerneis S, Kimmoun A, Klouche K, Lachâtre M, Lacombe K, Laine F, Lanoix J-P, Launay O, Laviolle B, Le Moing V, Le Pavec J, Le Tulzo Y, Le Turnier P, Lebeaux D, Lefevre B, Leroy S, Lescure F-X, Lessire H, Leveau B, Loubet P, Makinson A, Malvy D, Marquette C-H, Martin-Blondel G, Martinot M, Mayaux J, Mekontso-Dessap A, Meziani F, Mira J-P, Molina J-M, Monnet X, Mootien J, Mourvillier B, Murris-Espin M, Navellou J-C, Nseir S, Oulehri W, Peiffer-Smadja N, Perpoint T, Pialoux G, Pilmis B, Piriou V, Piroth L, Poissy J, Pourcher V, Quenot J-P, Raffi F, Reignier J, Revest M, Richard J-C, Riu-Poulenc B, Robert C, Roger P-A, Roger C, Rouveix-Nordon E, Ruch Y, Saidani N, Sayre N, Senneville E, Sotto A, Stefan F, Tacquard C, Terzi N, Textoris J, Thiery G, Timsit J-F, Tolsma V, Turmel J-M, Valour F, Wallet F, Wattecamps G, Yazdanpanah Y, Zerbib Y, Berna M, Reuter J, Staub T, Braz S, Ribeiro J-M, Paiva J-A, Roncon-Albuquerque R, Bouscambert-Duchamp M, Gaymard A, Lê M-P, Lina B, Peytavin G, Tubiana S, Couffin-Cadièrgues S, Esperou H, Belhadi D, Burdet C, Costagliola D, Dechanet A, Delmas C, Diallo A, Fougerou C, Guedj J, Mentré F, Mercier N, Noret M, Saillard J, Velou P. Remdesivir and three other drugs for hospitalised patients with COVID-19: final results of the WHO solidarity randomised trial and updated meta-analyses. Lancet. 2022;399(10339):1941-53. https://doi.org/10.1016/S0140-6736(22) 00519-0.
- Ader F, Bouscambert-Duchamp M, Hites M, et al. Remdesivir plus standard of care versus standard of care alone for the treatment of patients admitted to hospital with COVID-19 (DisCoVeRy): a phase 3, randomised, controlled, open-label trial. Lancet Infect Dis. 2022;22(2):209–21. https:// doi.org/10.1016/S1473-3099(21)00485-0.
- 37. Amstutz A, Speich B, Mentré F, et al. Effects of remdesivir in patients hospitalised with COVID-19: a systematic review and individual patient

data meta-analysis of randomised controlled trials. Lancet Respir Med. 2023;11(5):453–64. https://doi.org/10.1016/S2213-2600(22)00528-8.

- Mozaffari E, Chandak A, Zhang Z, et al. Remdesivir treatment in hospitalized patients with coronavirus disease 2019 (COVID-19): a comparative analysis of in-hospital all-cause mortality in a large multicenter observational cohort. Clin Infect Dis. 2022;75(1):e450–8. https://doi.org/10.1093/ cid/ciab875.
- Néant N, Lingas G, Le Hingrat Q, et al. Modeling SARS-CoV-2 viral kinetics and association with mortality in hospitalized patients from the French COVID cohort. Proc Natl Acad Sci. 2021;118(8):e2017962118. https://doi. org/10.1073/pnas.2017962118.
- Lingas G, Néant N, Gaymard A, et al. Effect of remdesivir on viral dynamics in COVID-19 hospitalized patients: a modelling analysis of the randomized, controlled, open-label DisCoVeRy trial. J Antimicrob Chemother. 2022;77(5):1404–12. https://doi.org/10.1093/jac/dkac048.
- Odeyemi YE, Chalmers SJ, Barreto EF, Jentzer JC, Gajic O, Yadav H. Early, biomarker-guided steroid dosing in COVID-19 *Pneumonia*: a pilot randomized controlled trial. Crit Care. 2022;26(1):9. https://doi.org/10.1186/ s13054-021-03873-2.
- Christensen, R, Weslay, A, Branscum, A B, Hanson, TE. Bayesian ideas and data analysis: an introduction for scientists and statisticians. Chapman&Hall/CRC. 2011.
- Demarqui FN, Loschi RH, Colosimo EA. Estimating the grid of time-points for the piecewise exponential model. Lifetime Data Anal. 2008;14(3):333– 56. https://doi.org/10.1007/s10985-008-9086-0.
- Helms J, Poissy J, Dequin PF, Timsit JF. Treatment of immunothrombosis dysregulation: high-dose corticosteroids is not the good option. Ann Intensive Care. 2023;13(1):58. https://doi.org/10.1186/ s13613-023-01158-1.
- 45. Ruberg SJ. Assessing and communicating heterogeneity of treatment effects for patient subpopulations: the hardest problem there is. Pharm Stat. 2021;20(5):939–44. https://doi.org/10.1002/pst.2110.

## **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

## Submit your manuscript to a SpringerOpen<sup>™</sup> journal and benefit from:

- Convenient online submission
- ► Rigorous peer review
- Open access: articles freely available online
- High visibility within the field
- Retaining the copyright to your article

Submit your next manuscript at > springeropen.com